Author’s response to reviews

Title: Acute Liver Failure Associated with Fructus Psoraleae: A Case Report and Literature Review

Authors:

ang li (frasting@163.com)

minhong gao (qyaominhong@163.com)

na zhao (269224131@qq.com)

ping li (472220440@qq.com)

jinguo zhu (1501229284@qq.com)

Wenfei li (lwf888026@163.com)

Version: 2 Date: 22 Sep 2018

Author’s response to reviews:

For Reviewer reports:

1. Suggest authors to weaken the definite conclusion or "patient dying from Fructus Psoraleae" throughout the whole manuscript, because this is authors' judgement. Since patient did not ONLY take Fructus Psoraleae, the causality of her liver failure and death could not be so certain. Suggest authors to use words such as "likely" to describe the causality.

   Thank you very much for your good suggestions. Yes, we have added “likely” before “due to” in line 130, page 5; in line 149, page 5; in line 159, page 6;

2. Language wise, further polishing and editing is still needed. There are some sentences are not written in the style of academic English in the current manuscript. I suggest authors working closely with a strong native editor of medical English to improve the quality of language.

   Thank you for your advice. Yes, we have revised language by native editor. For example: we have changed “Qubaibabuqi Pian” into “Qubaibabuqi Tablet” in manuscript. If you think there are still revisions in the language, please indicate the specific place.